Author's response to reviews

Title: Socio-Epidemiological Features of 2002 Plague Outbreak in Himachal Pradesh, India

Authors:

Sonu Goel (sonugoel007@gmail.com)
Harvinder Kaur (harvinderkaur315@gmail.com)
A K Gupta (anil1031@hotmail.com)
Umesh S Chauhan (uschauhan78@gmail.com)
Amarjeet Singh (amarminhas56@rediffmail.com)

Version: 2 Date: 20 March 2013

Author's response to reviews:

To
The
Editor-in-Chief
BMC Public Health
Dear Editor

Please find enclosed the revised version of the manuscript (MS: 1891864883917771) entitled, “Socio-Epidemiological Features of 2002 Plague Outbreak in Himachal Pradesh, India” with point-by-point description of the changes made for kind consideration of the editorial board of BMC Public Health. We thank you for careful and constructive review of our manuscript and the pertinent suggestions given to improve the aforesaid article. We think that the manuscript has been greatly improved by these revisions and we hope that you will now find it suitable for publication in BMC Public Health.

Point-wise response to the comments

1. Abstract has been structured according to the guidelines. Thanks for the pertinent suggestion. The requisite changes have been made.
2. Statement regarding the approval of the study by the ethics committee has now been included in the section ‘Material & Methods’
3. A 'Competing interests' section has now been incorporated in the manuscript
4. A section on 'Authors' contributions' has been added in the manuscript
5. An “Acknowledgments Section’ has been incorporated.
6. Methodology is now described in detail as suggested.
7. We have tried to improve the quality of written English of the manuscript and few changes have been made in the manuscript in general.

Thanking you with warm regards
Dr. Sonu Goel,
Assistant Professor of Health Management
School of Public Health, Department of Community Medicine, (PGIMER), Chandigarh, India
Phone: +91-172-2755215, 09914208490, 9914208027
Email: sonugoel007@yahoo.co.in